Author's response to reviews

Title: Survival in dialysis patients is not different between patients with diabetes as primary renal disease and patients with diabetes as a co-morbid condition.

Authors:

Marielle A Schriijen (m.a.schriijen@lumc.nl)
Olaf M Dekkers (o.m.dekkers@lumc.nl)
Diana C Grootendorst (D.C.Grootendorst@gmail.com)
Marlies Noordzij (m.noordzij@amc.uva.nl)
Johannes A Romijn (j.a.romijn@lumc.nl)
Raymond T Krediet (r.t.krediet@amc.uva.nl)
Elisabeth W Boeschoten (els@hansmakinstituut.nl)
Friedo W Dekker (f.w.dekker@lumc.nl)

Version: 4 Date: 1 December 2011

Author's response to reviews: see over
Dear Dr Chap,

We would like to thank you for accepting the article entitled ‘Survival in dialysis patients is not different between patients with diabetes as primary renal disease and patients with diabetes as a co-morbid condition.’

In our new version all the formatting changes had been made.

On behalf of all the authors,

Marielle Schriijen